Machine Supervisor last week

Machine Supervisor this week

* Beam commissioning in FTA and DI line:
- reference trajectory in FTA established with 1 nominal bunch

* Hardware commissioning in AD:

Summar - restart of the e-cooler
y - restart of stochastic cooling but IST not fully completed

- conditioning of the C10 cavity

* Beam commissioning in ELENA:

- analysis of scraper measurement

- problem with the FTA and DI BCT TRIC fixed on Monday

- problem with the magnetic horn IPOC triggering fixed
Issues - problem with the longitudinal movement of the target fixed

- QUAD-MAIN circuit regularly tripping
Plans Continuing HW activities in AD ring and beam commissioning of FTA/DI line
Yes / No Duration Preferred date/time
Reason
Impact




